

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Echeverría, Lucía; Gimenez-Nadal, J. Ignacio; Molina, José Alberto

Working Paper

Commuting in dual-earner households: International Gender Differences with Time Use Surveys

GLO Discussion Paper, No. 1307

Provided in Cooperation with:

Global Labor Organization (GLO)

Suggested Citation: Echeverría, Lucía; Gimenez-Nadal, J. Ignacio; Molina, José Alberto (2023): Commuting in dual-earner households: International Gender Differences with Time Use Surveys, GLO Discussion Paper, No. 1307, Global Labor Organization (GLO), Essen

This Version is available at: https://hdl.handle.net/10419/273425

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



Commuting in dual-earner households: International Gender Differences with Time Use Surveys*

Lucía Echeverría^{a,c}, J. Ignacio Gimenez-Nadal^{b,c} and José Alberto Molina^{b,c,d}

Abstract

Prior literature analyzing gender differences in commuting has reported that men commute longer distance/time than do women, and one explanation for this gender gap is based on household responsibilities falling on women. But most of the literature examining gender differences in commuting has not considered the interdependence that exists between the members of couples. We analyze gender differences in commuting time for a sample of dualearner couples living in Spain, Italy, South Korea, and the United Kingdom, taking into account the inter-relatedness of decisions within couples. We estimate Ordinary Least Squares equations for men and women on commuting time and mode of transport (private, public, and active transport) including own characteristics as well as spouse attributes and commuting choices. Results indicate that the number of children is significantly related to shorter commuting times for female workers in all countries, with no associations found for their male counterparts. In addition, having children is associated with changes in the commuting mode choice of women in Italy, Korea and the UK, but no associations are found for men. Our evidence indicates that, while the presence of children is related to commuting behavior of women, it is not the case for men. Furthermore, we find that couples' decisions on commuting are complementary, which may shed light on their relationship that should be addressed by theoretical models.

Keywords: commuting; gender differences, dual-earner couples; intra-spousal effects; household responsibilities; Multinational Time Use Study

JEL Classification: R40 - J22 - O57 - D19

^a CONICET, Argentina

^b University of Zaragoza and IEDIS, Spain

^c Global Labor Organization-GLO, The Netherlands

^d Institute of Labor Economics, IZA, Germany

^{*} This paper has benefitted from funding from the Spanish Ministry of Science and Innovation (Project PID2019-108348RA-I00), and from the Government of Aragón (Grant S32_20R).

1. Introduction

Commuting is an important part of the daily lives of workers worldwide, as millions of workers commute to their work daily. For instance, workers in the European Union spend an average of 25 minutes per day commuting (Eurostat, 2020), while workers in the United States spend around 47 minutes per day commuting (BLS, 2019). Given that these are significant portions of the 24 hours available in the day, and that there is a negative association between commuting and health-related outcomes of workers (Hansson et al. 2011; Künn-Nelen, 2016), the analysis of commuting behavior has received increasing attention in economics. Within the analysis of commuting behavior, a significant strand of the literature has centered on gender disparities. In general, men commute longer average distances and times compared to women. This is also observed in dual-earner households — a family arrangement that has received special attention over the years (Hanson and Hanson, 1980; Madden, 1981; Johnston-Anumonwo, 1992; Sultana, 2005).

Several reasons have been proposed to explain gender differences in commuting. Some studies have centered on the links between women's commuting and labor force participation (Madden, 1981; MacDonald, 1999), or on the geographical distribution of occupations (Hanson and Johnston, 1985; Hanson and Pratt, 1995; Benson, 2014). Other arguments state that gender differences in commuting behavior are closely related to the gendered division of housework and childcare. This latter explanation is known as the "household responsibility hypothesis" (Johnston-Anumonwo, 1992; Turner and Neimeier, 1997; Gimenez-Nadal and Molina, 2016), and posits that women have more space-time constraints, since they must adapt their out-of-home activities to their chores at home, which ultimately leads to shorter commuting times and distances (Turner and Neimeier, 1997; Lee and McDonald, 2003; McQuaid and Chen, 2012; Sandow and Westin, 2010; Gimenez-Nadal and Molina, 2016). Thus, household responsibilities may limit the commuting length of female workers.

⁻

¹ The negative health-related outcomes include lower subjective/psychological well-being (Dickerson et al. 2014; Roberts et al. 2011) and increased stress (Frey and Stutzer 2008; Gottholmseder et al. 2009; Novaco and Gonzalez 2009; Wener et al. 2003). These outcomes are important not only at the worker level, but also in terms of public health in general.

Household responsibilities may also affect the choice of the mode of transport used to commute. Women may be more likely than men to use public transit or active transport to go to work, even in situations when they have easy car access in the household (Matthies et al., 2002). However, greater household responsibilities may shift (i.e., increase) their priority to use the car to engage in household–support trips, given that women are more likely to engage in complex trip chains than men (Scheiner, 2014), and do most of the escorting of children (Motte-Baumvol et al., 2017). Scheiner and Holz-Rau (2012) study the intra-household allocation of cars, finding that having household responsibilities increases the chances of needing access to the car.

But most of the prior literature analyzing how household responsibilities are related to the commuting behavior of workers has focused on workers as single units, without taking into account the influence that other members of the household – mostly spouses – may exert on commuting behavior. Very few studies have explicitly incorporated this view in their methodological approach, since very few studies have analyzed the interdependence that exists between the decisions and outcomes of partners. When the relationship between the commuting behavior of spouses is taken into account, the evidence of the relationship to commuting between members of the same couple is not conclusive (Davis; 1993, Plaut, 2006; Picard et al., 2014; Roberts and Taylor, 2016; Chidambaram and Scheiner, 2020; Kwon and Akar, 2021; Oreffice and Sansone, 2022).

Within this framework, we analyze how male and female commuting duration is related to household responsibilities, in couples across countries, proxied by the number of children in the household. To that end, we use time use data from four developed countries – Spain, Italy, South Korea, and the United Kingdom – and explore the time devoted to commuting in dual-earner households.² In doing so, we also analyze the relationship of the commuting time of the members of the couple, given that prior evidence is inconclusive. Second, we explore how the mode of transport chosen for commuting by each member of the couple is related to household responsibilities, which is novel, as prior studies have not analyzed mode of transport decisions in the context of household responsibilities. A priori, women may be

_

² Because we focus on the daily commuting behavior of double-earner couples, we select those countries from the MTUS with available information on travels to/from work for both spouses.

more likely than men to use public transit due to their household responsibilities, or they have a lower priority to use the family car. Women with greater household responsibilities could be more dependent on a car due to their tight time budget and complex trip-chaining.

Because commuting decisions in couples are made jointly at the family level, we estimate Ordinary Least Squares (OLS) equations to model their decisions regarding commuting time and mode of travel (private, public, and active transport), in which the commuting decision of each spouse depends not only on own characteristics but also on the attributes and commuting behavior of his/her spouse. A key element of our methodological approach lies in the possibility of understanding whether commuting is complementary (i.e., trips to work are selected together as longer or shorter) or substitutable within the couple (i.e., one spouse tends to commute longer times, allowing the other spouse to commute shorter ones).

We find that having more children is significantly related to commuting times, but there are differences by gender and country. In Spain and Italy, female commuting time is shorter in the presence of children in the family (irrespective of how many), while in Korea and the UK it depends on the number of children. In contrast, having children is not associated with male commuting times. Then, it is expected that the gender gap in commuting time within couples widens as the number of children increases, supporting the household responsibility hypothesis across countries. Regarding mode choice, the presence of children is not related to changes in the male proportion of commuting time by any means of transport, but is associated with changes in the commuting mode choice of women in Italy, Korea and the UK (but not in Spain) even though results differ by mode of transport, number of children, and country. Our evidence also adds to prior studies showing that couples' decisions on commuting are complementary (Plaut, 2006; Roberts and Taylor, 2016), suggesting that couples jointly decide to increase (or decrease) their commuting duration as well as their fraction of commuting done by private, public, or active transport.

Our contribution to the literature is twofold. First, we add to the existing (but scarce) evidence on intra-spousal decision-making about commuting by examining the household responsibility hypothesis, while allowing for the dependence of commuting choices between spouses. We follow the standard approach of incorporating the number of children as an approximation of the levels of household responsibilities (Hanson and Johnston, 1985;

Johnston-Anumonwo, 1992; Lee and McDonald, 2003; McQuaid and Chen, 2012; Roberts and Taylor, 2016; Fan, 2017). This analysis is necessary, given that prior evidence of the relationship between commuting and household responsibilities is inconclusive. We also contribute to the analysis of the inter-dependence of commuting decisions within the couple, which few analyses have done in the past. Second, we extend the literature by exploring the links between the mode of transport used (private, public, and active) and household responsibilities. To our knowledge, no prior work has studied this while accounting for the interdependent nature of commuting decisions in couples.

The remainder of the paper is as follows. Section 2 presents a review of the literature. Section 3 presents the data and variables, Section 4 describes the empirical strategy, and Section 5 describes the results. Section 6 sets out our main conclusions.

2. Background

One significant strand of the literature has centered on the differences in the commuting behavior of men and women. The study of gender disparities in travel patterns has become important because it reveals how men and women decide on the allocation of responsibilities, their workplace, and their residential location. The literature shows that men commute longer distances/times compared to women, and different hypotheses have been proposed to explain this. Several authors have focused on the links between women's commuting and labor force participation (Madden, 1981; MacDonald, 1999). Some argue that because women tend to be the secondary wage earner within households, they work, on average, shorter hours and earn lower wages than men, increasing their costs of commuting (Waldfogel, 2007). Others argue for the importance of the geographical distribution of occupations, in the sense that differences in the configuration of home-work locations for women and men could explain why women engage in shorter commutes. Women are more likely to work in lower-status occupations and, if these jobs are less geographically concentrated than male jobs, then women probably work closer to their homes to reduce their commuting duration (Hanson and Johnston, 1985; Hanson and Pratt, 1995; Benson, 2014). However, gender differences in commuting decrease but do not disappear after controlling for income and occupation (Singell and Lillydahl, 1986; Hanson and Johnston, 1985; Sandow and Westin, 2010).

A different but central hypothesis involves gender roles and household responsibilities (Johnston-Anumonwo, 1992; Turner and Neimeier, 1997). The Household Responsibilities Hypothesis (HRH) states that gender differences in commuting behavior are closely related to the gendered division of household tasks. Evidence indicates that the burden of household work and childcare disproportionately lies on women (Aguiar and Hurst, 2007; Gimenez-Nadal and Sevilla, 2012). As a consequence, women have more space-time constraints, which affect their value of time (Rouwendal, 1999; Sermons and Koppelman, 2001; Brownstone and Small, 2005). They must adapt their out-of-home activities to their chores at home, which lead them to work in jobs closer to home and, ultimately, shorter commuting times and distances (Turner and Neimeier, 1997; Lee and McDonald, 2003; Sandow and Westin, 2010; McQuaid and Chen, 2012; Giménez-Nadal and Molina, 2016). Studies typically rely on household characteristics, such as marital status and the presence and number of children in the family, to approximate the levels of household responsibilities, and to test the hypothesis (Hanson and Johnston, 1985; Johnston-Anumonwo, 1992; Lee and McDonald, 2003; McQuaid and Chen, 2012; Fan, 2017). However, the evidence evaluating the extent of the gender differential in commuting patterns is mixed (Turner and Neimeier, 1997; Giménez-Nadal and Molina, 2016).

Part of the literature examining disparities in commuting patterns of men and women has placed special attention on the behavior of dual-earner households, because partners share a dwelling, but have separate working places. Commuting is central to work-home arrangements because it is the nexus between the worker and the housing market (Roberts and Taylor, 2016). Evidence of work trips in dual-earner families also indicates that women commute shorter distances and shorter times (Hanson and Hanson, 1980; Madden, 1981; Johnston-Anumonwo, 1992; Sultana, 2005), but men and women are affected differentially by household characteristics, household type, and housing tenure (Kim, 1994, 1995). Additionally, the literature has focused on comparing commuting decisions of dual-earners to those of single-earners (Green, 1997; McQuaid and Chen, 2012; Hirte and Illmann, 2019). Findings suggest that dual-earner households are more likely to live farther from their work locations than single-earners (Madden, 1980). However, workers of dual-earner households commute, on average, the same, or even less, than individuals in single-earner households (Rouwendal and Rietvald, 1994; Sultana, 2005; Surprenant-Legault et al., 2013), probably

because the home location could be chosen to minimize joint commuting distances (Kim, 1995), despite facing more constraints in balancing home and work locations.

These studies have analyzed different aspects of the gender disparities in commuting behavior by comparing different family types at the household-level (single- and double-earners), or by focusing separately on male and female individual decision-making in the context of dual-earner households. However, little work has been done on examining gender commuting differences by taking into account the interdependence that exists between the decisions and outcomes of partners in dual-earner families. Couples reach joint decisions in a variety of ways, as a result of interactions and bargaining processes. Partners make a joint decision regarding the location of their home, but make separate but dependent choices on employment location, which ultimately determines each spouse's commuting time (Roberts and Taylor, 2016).

Few studies explicitly account for the dependent nature of commuting decisions in dual-earner families. Plaut (2006) focuses on the role of housing in the inter-relationship between male and female commuting decisions of spouses in dual-income households in the US, estimating a Seemingly Unrelated Regression (SUR) model to account for the interdependence of commuting decisions between partners, and finding that men commute further than women, and owners commute further than renters. Following the same methodological approach, Roberts and Taylor (2016) introduce the conditions of local labor markets to analyze spouses' commuting behavior in the UK, finding that men's commuting times are more sensitive to local unemployment rates. Plaut (2006) and Roberts and Taylor (2016) find that commuting by men and women are complements rather than substitutes, meaning that work trips are adjusted together to be longer (or shorter) for both spouses. The complementary nature of spouses' commuting may be interpreted as a result of jointly made and mutually beneficial commuting decisions in order to obtain better housing and location opportunities (Plaut, 2006). Consequently, women's shorter commutes may not reflect their relatively weaker positions and lower bargaining power within the household.

However, Surprenant-Legault et al. (2013) argue that this complementarity does not imply that couples do not engage in trade-offs, but rather that they apply strategies in order to decrease their total commuting distance. Further, the evidence of complementary commuting

behavior within couples in dual-earner families is not conclusive, since earlier findings for the US suggest that commuting in couples is a substitute, in the sense that one spouse spends more time travelling to/from work in exchange for shorter trips for the partner (Davis, 1993). In turn, Kwon and Akar (2021) analyze the determinants of the household total commuting distance and share of women's commuting distance in the US, and find that commuting mode, the presence of children, and occupation-related characteristics affect the gender gaps within couples.³

But despite that this prior literature has developed several intra-household decision models, the models ignore the bargaining process within the family, as studied by family economists (Picard et al. 2014). One exception to this argument is Oreffice and Sansone (2022), who develop a model to show how to non-cooperatively allocate spouse's time between work in the labor market and household production, as in Bertrand et al. (2021), augmented with commuting time decisions.

3. Data and Variables

Our analysis relies on the Multinational Time Use Study (MTUS) data set, coordinated by the Centre for Time Use Research (CTUR) at University College, London, and incorporated in the Integrated Public Use Microdata Series (IPUMS) of the Institute for Social Research and Data Innovation of the University of Minnesota (Fisher et al., 2019). The MTUS is a dataset aimed at harmonizing time-use surveys worldwide and includes information on 69 activities performed by individuals during the day, for randomly selected samples, from 25 countries over 5 decades, including travel activities. In addition, the MTUS collects information on individual and family-level socio-demographics. Data is gathered via completion of personal diaries as well as individual and household questionnaires.

There is a growing literature using time-use surveys to analyze transportation behavior (Gimenez-Nadal and Molina, 2014; 2016; Jara-Díaz and Rosales-Salas, 2015; Gimenez-

_

³ Using a different methodological approach, Chidambaram and Scheiner (2020) examine the gender gap in commuting distance within dual-earner couples in Germany. They find that gender disparities in economic prospects increase the gender gap in commuting distances. In addition, their evidence indicates that a relative dominance of car access by the female spouse reduces the gap, while an increase in time spent on unpaid work by the male partner decreases the gender gap.

Nadal et al., 2018a, 2018b, 2022a; Echeverria et al., 2022). Such surveys have become the "gold standard" to study other uses of time, including paid work, housework, and childcare (Aguiar and Hurst, 2007; Gimenez-Nadal and Sevilla, 2012), since the analysis derived from time use data yields more reliable and accurate estimates in comparison to time use information obtained from stylized questions (Robinson and Godbey 1985; Juster and Stafford 1985).

We select those countries from the MTUS with available information on travel to/from work for both members of the dual-earner couple.⁴ The MTUS currently includes four developed countries whose surveys include that information, for the 2000s and 2010s: Spain (2002-2003 and 2009-2010), Italy (2002-2003 and 2008-2009), South Korea (2009), and the United Kingdom (2000-2001 and 2014-2015).⁵ We restrict the analysis to heterosexual working couples aged between 21 and 65 years old (Aguiar and Hurst, 2007, Gimenez-Nadal and Sevilla, 2012) and with both members reporting positive time devoted to commuting during working days (Gimenez-Nadal and Molina, 2019; Gimenez-Nadal et al., 2018a, 2018b; Molina et al., 2020). The final sample is 6,751 couples (see Table A.1 for sample composition by country).

Our analysis focuses on two sets of dependent variables. First, we are interested in the commuting times of each member of the couple, defined as the minutes per day that they devote to travel to/from work (i.e., *Commuting time*). Second, we are interested in the proportion of commuting time by private transport (car, truck, or motorbike), public transit, and active transport (walking and cycling). To construct these variables, defined at the individual level, we sum for each individual the commuting time (in minutes) by each mode

-

⁴ Following Browning et al., 1994, Blundell et al., 2005; Cherchye et al., 2012, we focus on dual-earner households, so that participation decisions in the labour market are not analyzed. The analysis of single-earner households may lead to different conclusions, and raises concerns about selectivity of couples. As commuting is a function of work, those who do not work do not commute, and perhaps they do not work because they do not want to commute. Those who work may be innately different than those not working, especially given that wives who were at one point out of the labor force (perhaps because of children) eventually re-enter the labor force. Furthermore, those dual-earner couples where any of the members do not commute are excluded from the analysis, leaving aside the analysis of teleworkers. Despite that telework (e.g., work from home) has become a common practice today, for the years on which the surveys are based this practice was not as widespread, and thus this issue may have little effect on the conclusions drawn here.

⁵ We do not consider time use surveys in previous decades (e.g., 1980s, 1990s) because we want to give an up-to-date view of commuting behavior within couples.

⁶ Because information on travel distance is not available in the MTUS, we rely on information regarding individual commuting times.

of transport, and divide it by the total time spent in commuting during the day by the individual.

Table 1 shows the time devoted to daily commuting and the proportion of that time done by private, public, and active modes of transport, by country and gender. We also report the results from statistical mean tests of gender differences (t-type test of equality of means, where H0 refers to the equality of means for men and women). We observe that, on average, men commute for longer times than do women, in all countries of the sample. The largest differences in commuting times between men and women are found for the UK (17.9 minutes), followed by Italy (8.6 minutes), Spain (7.1 minutes) and Korea (4.6 minutes). These differences are statistically significant at the 1% level, indicating that men in double-earner couples commute about 37% more than their female partners in the UK, 16% more in Italy, 14% more in Spain, and 8% more in Korea.

Regarding the mode of transport used for commuting, we observe important and cross-country-consistent differences by gender. For all countries of our sample, men spend on average a larger proportion of their commuting time using private transport (ranging from 73.5% in Spain to 79.8% in the UK), while women travel a larger fraction by public transit (ranging from 6.4% in Italy to 14.1% in Spain) and active transport (ranging from 20% in Italy and the UK to 41.8% in Korea). Further, gender differences are larger in the proportion of commuting by private car, followed by differences in active transport, while gender differences in the use of public transit are much smaller. In turn, these differences are consistently larger for Korea, followed by Spain, Italy, and the UK, irrespective of the mode of transport. Mean differences are statistically significant at the 1% level, except for the proportion of commuting time by public transport in the UK.

To account for the observed heterogeneity across individuals and couples, we include individual and household characteristics, including the number of children in the family. For personal characteristics, we consider age and the highest level of formal education achieved (primary education/uncompleted secondary, completed secondary, and higher education). Roberts and Taylor (2016) find that younger, more educated men and women in the UK commute longer, but the gradient is steeper for men (women) in the case of education (age). We also incorporate the hours of paid work per week and occupational category (Schwanen

and Dijst, 2002; Gutierrez-i-Puigarnau and van Ommeren, 2010; Gimenez-Nadal and Molina, 2014), which has been found to be significantly related to the commuting distance gap in German couples (Chidambaram and Scheiner, 2020).

We include the number of children under age 18 in the household, to proxy for household responsibilities, following Hanson and Johnston (1985), Johnston-Anumonwo (1992), Lee and McDonald (2003), McQuaid and Chen (2012) and Roberts and Taylor (2016). Children and childcare activities may impose differential constraints on the commuting of partners, and are affected by the opportunity cost of time. Evidence of the relationship between the presence of children and commuting is mixed. Some studies find that having a child is not a significant factor in commuting distance for men or women (Kim, 1994, 1995). In addition, having children is not related to the gender gap in commuting distance in the US (Chidambaram and Scheiner, 2020). In contrast, one study for the US suggests that increases in the number of children are likely to lead to longer commuting (total) distances for two-earner couples, while likely to decrease the share of women's commuting distance in total household commuting (Kwon and Akar, 2021). In the case of the UK, having children bears no relationship to men's commuting times but is associated with shorter commuting time for women (Roberts and Taylor, 2016).

We include a set of household characteristics, such as the residential location (urban/suburban or rural/semi-rural), ownership of a home (own outright, mortgage, or rent) and ownership of at least one motorized vehicle (either car or motorcycle)⁷. Abraham et al. (2010) find a differential willingness between the male and female to change the residential location in response to work-related incentives, and that a bargaining process operates within couples to minimize potential conflict resulting from migration. In this line, Mok (2007) argues that when there are children in the family, it is necessary to account for family-decision making in location decisions. In addition, evidence for the US (Plaut, 2006) and the UK (Roberts and Taylor, 2016) shows that owners commute further than renters, reflecting rigidities in the housing market. Car ownership is associated with shorter commuting times for women in the case of renters, and to longer trips for men homeowners in the US (Plaut,

⁷ Information on ownership of a home and motorized vehicles is not available for Spain (2009-2010); ownership of motorized vehicles is not available for Korea, while urban location is not available for the UK (2014-2015).

2006), while the number of cars in the household is associated with shorter commuting times for women in the UK (Roberts and Taylor, 2016). However, in terms of the gender gap in commuting between partners in Germany, car availability is not related to commuting distance. In addition, there is no indication of greater gender equality in commuting in urban areas (Chidambaram and Scheiner, 2020).

Panel (A) of Table 2 reports summary statistics of the socio-demographic characteristics by gender. Men are, on average, 42.7 to 45.8 years old, depending on the country, and are slightly older than women (40.6 to 42.8 years old). There is a larger proportion of men (women) with primary (or uncompleted secondary) education in Spain, Italy and the UK (Korea). In Spain (Italy and Korea), the proportion of men (women) with secondary education is larger, while in the UK it is of the same order. In Spain, Italy, and the UK (Korea), there is a larger proportion of women (men) with higher education. In all countries, the proportion of men and women who only achieved a primary educational level is the lowest. In Spain (Italy) there is a larger fraction of individuals with higher (secondary) education, while the distribution of individuals across secondary and higher levels of education is quite similar in the UK. Men work more hours per week than do women. Gender differences in paid work are the greatest in the UK (11.6 more hours per week), followed by Korea (7.1 hours), and Spain (5.9 hours).

Panel (B) of Table 2 reports the proportion of couples without children and with one, two, or more than two children. In Spain, Italy, and Korea, approximately 40% of couples do not have children, while that figure is 50% in the case of the UK. Spain and Italy have a larger proportion of couples with one child (31% and 34%, respectively) than with two children (26% and 23%, respectively). Korea and the UK have a larger proportion of couples with two children (35% and 24%, respectively) than with one child (21% for both countries). In all countries, the percentage of couples with more than two children is comparatively low (between 4% and 6%).

Panel (C) of Table 2 shows that Italy has a lower proportion of couples living in urban/suburban areas (62%), followed by Spain (72%) and Korea (94%). At the same time, 68% of couples in the Korean sample are home-owners, while this number is 78% in Italy

⁸ Information on working hours is not available for Italy (2002-2003).

and 84% in the UK. In addition, almost all Italian couples in the sample own at least one motorized vehicle (99%), while only 60% of couples in the UK are owners of a car or a motorcycle.

4. Empirical Strategy

We analyze the decisions of spouses regarding commuting times and mode of transport in dual-earner couples, focusing on gender and cross-country differences. In our analysis, we take into account that commuting decisions of the members of couples are inter-related. To model this interdependence in commuting decisions, we estimate Ordinary Least Squares (OLS) regressions separately for the male and the female, including in each equation not only own characteristics but also the attributes and commuting behavior of the spouse.

First, we estimate an OLS model in which the dependent variables are the male and female daily commuting times in minutes (CT), and these variables are transformed to log form to interpret estimated coefficients directly as (semi) elasticities (Plaut, 2006; Roberts and Taylor, 2016). Model (1) is composed of two separate equations (Eq. (1.1) and (1.2)), one for each spouse (m = male and f = female) living in country (c), specified as follows:

$$\log\left(CT_{c}^{m}\right) = \alpha_{c}^{m} + \beta_{c}^{m}X_{c}^{m} + \gamma_{c}^{m}X_{c}^{f} + \theta_{c}^{m}CT_{c}^{f} + \eta_{c}^{m}CH_{c} + \delta_{c}^{m}H_{c} + \lambda_{c}^{m}FE_{c} + \varepsilon_{c}^{m}$$
(1.1)

$$\log\left(CT_{c}^{f}\right) = \alpha_{c}^{f} + \beta_{c}^{f}X_{c}^{f} + \gamma_{c}^{f}X_{c}^{m} + \theta_{c}^{f}CT_{c}^{m} + \eta_{c}^{f}CH_{c} + \delta_{c}^{f}H_{c} + \lambda_{c}^{f}FE_{c} + \varepsilon_{c}^{f}$$
(1.2)

where X_c^m and X_c^f are vectors containing the socio-demographic variables for the male and female, respectively. These include age (and its square), the highest level of formal education achieved (elementary, secondary or higher education), and the number of hours of paid work per week. CH_c is a vector of indicator variables at the household-level of the number of children under age 18. Specifically, we include an indicator for couples with one child, two children, and more than two children (couples with no child are the reference category). 9H_c

⁹We have alternatively included the age of the youngest child to assess how children's age relates to commuting time, as children's age may have an important bearing on household responsibilities and hence on commuter choices. This information is only available for Italy and South Korea. Results indicate that an older youngest child is significantly related to shorter female commuting times in the case of Italy (at 10% level), with no

includes a set of household characteristics, such as the residential location (urban/suburban or rural/semi-rural), ownership of a home (own outright, mortgage or rent) and ownership of at least one motorized vehicle (either car or motorcycle). In the vector FE_c we include control variables to account for the occupational category of the men and women, and the region and year of the survey, when available. Standard errors are robust.

Our focus is on two sets of parameters. On the one hand, we are interested in η_c^m and η_c^f , as they show how male and female commuting duration is related to household responsibilities in couples across countries, proxied by the number of children in the household. A positive (negative) correlation for a given spouse would provide evidence that having more children is related to longer (shorter) commuting times of that spouse. On the other hand, we are also interested in assessing the sign of θ_c^m and θ_c^f , which associates the commuting time of one spouse with the commuting time of the other spouse. A significant association would indicate that commuting decisions in dual earner-couples are not independent. Further, a positive correlation would indicate that spousal commuting is a complementary, in the sense that trips to work are selected together as longer or shorter, conditional on individual characteristics and living arrangements. In contrast, a negative correlation suggests that spousal commuting is substitutable, meaning that one spouse tends to commute for longer trips to work, allowing the other spouse to commute for shorter ones. ¹⁰

Second, we estimate model (2) composed of Eq. (2.1) and (2.2) to analyze how the mode of transport chosen to commute by each member of the couple is related to household responsibilities. We consider three alternative dependent variables in the SUR regressions: a) the proportion of commuting time done by private transport, b) the proportion of commuting time done by public transit, and c) the proportion of commuting time done by an active mode of transport. We estimate the following specification for each country, including the same control variables as in Eq. (1.1) and (1.2):

-

association found for male commuting time. In contrast, an older youngest child is significantly related to longer female and male commuting times in the case of South Korea. Results are available upon request.

¹⁰ In alternative regressions, we assess if bargaining power between spouses, captured by the difference in education level (Lundberg and Pollack, 2008), mediates the relationship between the number of children and commuting choices, by including an interaction variable. However, results are, in general, not statistically significant. Results are available upon request.

$$\log(P_c^m + 1) = \alpha_c^m + \beta_c^m X_c^m + \gamma_c^m X_c^f + \theta_c^m P_c^f + \eta_c^m C H_c + \delta_c^m H_c + \lambda_c^m F E_c + \varepsilon_c^m$$
 (2.1)

$$\log\left(P_c^f + 1\right) = \alpha_c^f + \beta_c^f X_c^f + \gamma_c^f X_c^m + \theta_c^f P_c^m + \eta_c^f C H_c + \delta_c^f H_c + \lambda_c^f F E_c + \varepsilon_c^f \tag{2.2}$$

where P_c^m and P_c^f is the proportion of travel time by private, public transit, or an active mode by the male "m" and the female "f" in country "c", respectively. We also transform dependent variables to their logarithm form to interpret the estimated coefficients as elasticities. Similarly to Eq. (1.1) and (1.2), we allow for the commuting choice mode of each spouse to depend on the choice of his/her partner.

5. Results

5.1. Commuting time

We present the results of estimating models (1) and (2) for dual-earner couples who travel to/from work in Spain, Italy, Korea and the UK. Regressions modelling commuting choices are separately estimated by gender and by each country in our sample. Estimated coefficients are interpreted as changes in percentage levels of the male and female commuting time.

Table 3 reports the estimates of model (1). We find that having more children is significantly related to commuting time, but there are differences by gender and country. In Spain and Italy, female commuting time is, on average, shorter when there are children in the family, irrespective of the number of children (and reductions are larger for an increasing number of children), while in Korea and the UK it depends on the number of children. Specifically, in Spain, having one child is associated with a 12.2% decrease in female commuting time with respect to women without children, while having two children is related to a 22.6% decrease, and having more than two children to a 27.4% decrease in time spent travelling to/from work. A similar result is found for Italy, where decreases in female commuting time are 11.2%, 21.8%, and 32.5%, respectively. In the UK, reductions of female commuting time are found in couples with one child (15.2%) and two children (19.2%), while no association is found in the case of having more than two children. In Korea, there is a

-

¹¹ We sum 1 to our dependent variables to avoid problems computing logarithms for individuals reporting no travel time by public or active transport.

reduction of female commuting time in couples with more than two children (14.6%), in comparison with childless couples, but no association is found for women with only one or two children.

We find that the presence of children is not associated with male commuting times in any of the four countries, with the exception f Italian couples with more than two children, for which we observe an increase in commuting time of 15.8% relative to men in childless couples. Given the lack of correlation between male commuting time and the number of children, it is expected that the gender gap in commuting time within couples widens as the number of children increases. These results are evidence for the household responsibility hypothesis across countries.

We find few significant associations between individual and family-level characteristics and commuting time. Regarding own characteristics, we find that more educated men commute longer in the UK but shorter in Spain, while more educated women commute longer in Italy and the UK. In turn, men working more hours per week commute significantly shorter times only in Korea, while women working more hours per week commute significantly longer times in Spain and the UK. The characteristics of the spouse are significantly correlated with own commuting time in the case of Spain. Specifically, when men (women) work more hours per week, women (men) commute less time, while in families with relatively older women we observe that male commuting time is longer. Regarding family-level attributes, we find that living in an urban/suburban location is positively associated only with male commuting time in Korea, while being the owner of a motorized vehicle is related to shorter female commuting time in the UK.¹²

Furthermore, our results provide evidence of a positive and statistically significant association between spouses' commuting time for all countries. The greatest positive association is found for Korea, while the lowest one is for the UK. That is, spouses increase (or reduce) their commuting times together, after controlling for individual and family characteristics. These results are in line with prior evidence using SUR estimates to analyze

15

¹² In alternative specifications we include a set of indicators to account for the number of vehicles. However, these variables are not significant.

commuting distances and times, in dual-earner couples for the UK (Roberts and Taylor, 2016) and the US (Plaut, 2006).

Several explanations are possible for this positive relationship between the commuting time of the spouses. The first is based on the assortative mating by education observed in marriage markets (Chiappori et al., 2020a, 2020b; Eika et al., 2019). To the extent that spouses match with partners of similar educational levels, highly-educated male workers tend to match to highly-educated female workers. On the other hand, higher education is related to higher wages (Heckman et al., 2006; Carneiro et al., 2011; Gunderson and Oreopolous, 2020), which creates two contrary effects on the job search market: 1) higher wages imply higher opportunity costs of commuting, and 2) higher wages imply longer commuting to reach those job places. The evidence is positive regarding the relationship between commuting and wages (Gimenez-Nadal and Molina, 2014; Gimenez-Nadal, Molina and Velilla, 2018b), indicating that the second effect dominates, and could explain why the relationship between the commuting for the members of the couples is positively related in highly-educated couples. In an intra-household context, in which wages are not controlled for, the fact that there is this complementarity means that the preferences of the spouses for a greater degree of specialization (higher wages) overcome the opportunity cost generated by commuting. In other words, if one of the spouses travels a lot to get a good salary, he/she prefers that his/her partner does the same and also travels to get a better salary.

Another explanation could be based on togetherness and synchronization with spouse's leisure. If one of the partners travels a lot (which reduces his/her leisure time, Gimenez-Nadal, Molina and Velilla (2018b, 2021), he/she has less time available to spend leisure time with his/her partner. But if his/her partner travels little, she/he will have leisure time that she/he will not be able to share with the spouse, but alone, which will provide less utility than the leisure time shared with the partner (Hamermesh, 2002,2020; Hallberg, 2003; Jenkins and Osberg, 2004; Cosaert et al., 2023; Gimenez-Nadal, Molina and Velilla, 2023). On the contrary, if one of the partners travels very little, he/she will have more time available to spend leisure time with his/her partner. But if the partner travels a lot, he/she will have leisure time that he/she will not be able to share with the spouse, but alone, which will provide

less utility. Thus, there may be a preference for couples to be together for leisure, creating a positive relationship between the commuting time of the spouses.

5.2. Commuting modes

Tables 4, 5, and 6 report the results of estimating model (2) for the proportion of male and female commuting time by private transport, public transit, and active transport, respectively, by gender and country. In Table 4, we observe that Italian women spend on average, a significantly larger proportion of their commuting time by car when there are one or two children (4.2% and 4.7% more than childless women), while in the UK only women living in couples with more than two children spend a larger proportion of their commuting time by car (7.2%). Table 5 shows that the fraction of commuting by public transport declines, on average, for Spanish men with two children (2.4%), for Italian women with one child (1.6%), and two children (3.1%), for Korean women with two (4.0%) and more than two (5.4%) children, and for women in the UK with more than two children (3.6%). Table 6 indicates that only Italian women with one child experience an average reduction in the fraction of active commuting of 2.5%.

In sum, we find that, after conditioning for individual and household characteristics, the presence of children is not related to the male proportion of commuting time by car, public and active transport in any country (with the exception of commuting by public transit in Spain when couples have two children). However, having children is related to changes in the commuting mode choice of women in Italy, Korea, and the UK. Specifically, it increases the proportion of female time commuting by private transport in Italy (one or two children) and in the UK (more than two children); decreases the fraction of female commuting by public transit in Italy (one or two children), Korea (two or more than two children) and the UK (more than two children), while decreasing the fraction of female active commuting in Italy (one child). In particular, Italian women seem to change their commuting mode of transport the most in the presence of children, followed by English and Korean women with several children. In contrast, Spanish women do not significantly alter the fraction of time commuting by each mode of transport in the presence of children.

Decisions regarding mode of transport are inter-connected between spouses, as captured by the positive sign of the commuting decision of one spouse on the commuting choice of the other spouse. As in estimates of model (1), the strongest associations are found for Korean couples. This evidence indicates that partners jointly decide to increase (or decrease) their fraction of commuting by private, public, and active transport together, rather than substituting modes of transport. This could also be a matter of sorting of couples, as couples more concerned with environmental issues may be using a higher proportion of alternative modes of transport to car use. This topic may be interesting for future research.

Several individual and household characteristics relate to the gendered use of modes of transport but differ by country. In Table 4, we observe that own age is significantly and positively associated with both the male and female fractions of commuting time by private transport in Korea. Women with higher education commute proportionally more by car than do less educated women in Korea. Women working more hours per week also commute a larger fraction of their time by private transport in Spain, Korea, and the UK, while this fraction also increases for men working more hours in the UK. Regarding the relationship between the characteristics of the spouse and commuting choices of the other spouse, we find that the male commute is longer (shorter) if the spouse has secondary (higher) education in the UK (Korea). In turn, female commuting is longer if the spouse has secondary (higher) education in Italy (the UK).

Men and women in urban locations spend a smaller proportion of their commuting time by private transport in Spain, but a larger change is found for women (7.2% vs. 3.0%). Homeowners spend a larger fraction of their commuting by private transport, but this association is more pronounced for women in Italy and the UK, while it is of the same order as partners in Korea. Having a motorized vehicle naturally increases the proportion of commuting time by private transport (in Italy and the UK), but the correlation is stronger for men. This result is in line with prior evidence suggesting that women have less access to a car in households sharing a car (Scheiner and Holz-Rau, 2012).

For public transit (Table 5), age is significantly and positively associated with men's fraction of commuting by private transport in Italy and the UK. Having a higher education decreases the fraction of commuting by public transit for women in Korea. Working more

hours per week is significantly and negatively associated with public transit for women in Korea and for men in Spain and the UK, but positively associated for women in the UK. In the UK, in couples where the man has secondary or higher education, or works more hours per week, the proportion of female commuting by public transit decreases, while the male proportion increases when the spouse has a higher education. Living in urban areas is positively and significantly associated with the fraction of commuting by public transit for both men and women in Spain, with a larger estimate for women, but only for women in Italy and men in Korea. Being a homeowner decreases the fraction of commuting time by public transit for men (women) in Italy (the UK), but is not associated with their female (male) counterparts. Having a motorized vehicle decreases the proportion of commuting time by public transit only for men in Italy and the UK.

Regarding the proportion of active commuting (Table 6), own age is positively (negatively) and significantly related to women's proportion in the UK (Korea), and negatively related to men's in the UK and Korea. Having a higher education is negatively and significantly associated with the proportion of active commuting for women in Korea. Working more hours per week is negatively and significantly associated with women's proportion of active commuting in Spain and the UK, but this correlation is positive for men in Korea. In turn, male (female) age is positively (negatively) related to women's (men's) proportion in Korea (Spain). In addition, when men (women) have a secondary education, the fraction of active commuting of the spouse decreases in Italy (the UK), and when men are older the female proportion increases in Korea. Living in urban areas is only significantly associated with the fraction of male active commuting time in Korea. Being a homeowner decreases, on average, the fraction of active commuting for both spouses in Korea and the UK, but only for women in Italy. Having a motorized vehicle significantly decreases the proportion of active commuting for both spouses in Italy and the UK, but the correlation is stronger for women in the UK, and somewhat similar in Italy.

In order to test the robustness of the results from Equation (2), we also estimate a multinomial logit to account for the fact that the choice of any mode of transport is dependent on the choice of the other two modes. Furthermore, adding 1 to each proportion of transport mode implies no discontinuity between zero use of a mode and a tiny use of a mode, but there

must still be a substantial discontinuity at 0 for mode choice. Results (reported) in Table 7) are robust in comparison to estimates of model (2) (reported in Tables 4 to 6), supporting our main conclusions of the relationship between mode choice by gender and number of children in the family across countries.

6. Conclusions

Prior literature analyzing gender differences in commuting reports that men commute longer distance/time than do women, and one explanation for this gender gap is that the bulk of household responsibilities falls to women. However, most of the literature examining gender differences in commuting has not considered the interdependence that exists between the members of couples. In this paper, we analyze how male and female commuting duration is related to household responsibilities, in couples from four developed countries, allowing for the possibility that one spouse's decision on commuting is related to that of the other. We find that having more children is significantly related to commuting times, but there are differences by gender and country. In Spain and Italy, female commuting time is shorter when there are children in the family (irrespective of how many), while in Korea and the UK this depends on the number of children. In contrast, having children is not associated with male commuting times. Then, it is expected that the gender gap in commuting time within couples widens as the number of children increases, supporting the household responsibility hypothesis across countries. Regarding commuting mode choice, the presence of children is not related to changes in the male proportion of commuting time by any means of transport, but is associated with changes in the mode choice of women in Italy, Korea, and the UK (but not in Spain) even though results differ by mode of transport, number of children, and country.

Our results may serve to stimulate further research on the topic of commuting behavior and its connection to household responsibilities. Theoretical, as well as further empirical, research is needed to shed light on the question of how gender affects individual commuting behavior. Furthermore, employment policies should consider the relationship between commuting and household responsibilities, since more family-friendly policies may increase the desire of women to work farther from home, ultimately increasing their labor force

participation – at least in the countries where we find evidence of limitations due to household responsibilities.

Identifying which workers are more likely to use public and/or active transport for commuting is important for firms and policy makers. For instance, the fact that the mode of transport is affected by household responsibilities may indicate that children interact with patterns of use of sustainable commuting (e.g., the use of green modes of transport, such as public or active transport), and thus incentives to those who are less likely to use these services may help to increase the use of such modes. This may include, for instance, offering discounts to working parents when using Mobility As a Service (Maas) applications, offering discounts to those working parents who use public transit, or offering free public transit services to children under a certain age. Far from being exhaustive, this paper does not offer a complete review of the existing incentives, and further research could focus on whether such measures are effective in increasing the use of public transport or bike-sharing services for working parents.

The fact that we find complementarities between the commuting time of the members of couples is at odds with common hypotheses and with findings in much of the prior literature, which may have implications at the theoretical level. Job search models (van den Berg and Gorter, 1997; van Ommeren, 1998; Rouwendal, 2004) consider commuting as a source of labor mobility that allows workers to access geographically dispersed labor markets without the need to migrate (Cameron and Muellbauer, 1998). In the field of transport economics (DeSalvo and Huq, 1996), commuters choose a mode of transportation to minimize the monetary and opportunity costs of travel. Theoretical models in the field of Urban Economics focus on the location of the home, where displacements are generally assumed to result in disutility, and households are located to maximize the utility obtained from the dwelling and all other goods (e.g., the monocentric city model, as in Alonso (1964), Mills (1967), and Muth (1969), and the "Polycentric City Model" developed by Muller (1981), Garreau (1991), and Knox and McCarthy (2005)). The fact that we find complementarities in commuting between the members of the couple may serve to corroborate or further develop these models.

Our analysis and results present several limitations. First, given the cross-sectional nature of the data, the results cannot be interpreted as causal, since time-variant and time-invariant

factors at both the individual and household level may be biasing the observed results. Furthermore, our definition of commuting time is restricted to commuting episodes only, and no chained trips (e.g., non-commuting trips while commuting to or from work) are included in the commuting journeys. This represents a limitation, as conclusions can change if a wider definition of commuting is used, especially when it comes to gender differences in commuting time (Gimenez-Nadal, Molina and Velilla, 2022b). More research on this topic is needed.

REFERENCES

- Abraham, M., Auspurg, K., and Hinz, T. 2010. Migration decisions within dual-earner partnerships: A test of bargaining theory. *Journal of Marriage and Family*, 72(4), 876-892.
- Aguiar, M. and Hurst, E. 2007. Measuring Trends in Leisure: The Allocation of Time Over Five decades, *Quarterly Journal of Economics*, 122(3), 969–1007.
- Alonso, W., 1964. Location and Land Use. Harvard University Press, Cambridge
- Benson, A. 2014. Rethinking the two-body problem: The segregation of women into geographically dispersed occupations. *Demography*, 51(5), 1619-1639.
- Bertrand, M., Cortes, P., Olivetti, C., and Pan, J. 2021. Social norms, labour market opportunities, and the marriage gap between skilled and unskilled women. *The Review of Economic Studies*, 88(4), 1936-1978.
- Blundell, R., Chiappori, P. A., and Meghir, C. 2005. Collective labor supply with children. *Journal of political Economy*, 113(6), 1277-1306.
- Browning, M., Bourguignon, F., Chiappori, P. A., and Lechene, V. 1994. Income and outcomes: A structural model of intrahousehold allocation. *Journal of political Economy*, 102(6), 1067-1096.
- Brownstone, D. and Small, K.A. 2005 Valuing time and reliability: assessing the evidence from road pricing demonstrations. *Transportation Research A: Policy and Practice*, 39, 279–93.

- Bureau of Labor Statistics. (2019). American Time Use Survey News Release. U.S. Department of Labor
- Cameron, G., and Muellbauer, J. 1998. *The housing market and regional commuting and migration choices* (No. 1945). CEPR Discussion Papers.
- Carneiro, P., Heckman, J. J., and Vytlacil, E. J. 2011. Estimating marginal returns to education. *American Economic Review*, 101(6), 2754-2781.
- Cherchye, L., Rock, B. D., and Vermeulen, F. 2012. Married with children: A collective labor supply model with detailed time use and intrahousehold expenditure information. *American Economic Review*, 102(7), 3377-3405.
- Chiappori, P. A., Costa-Dias, M., and Meghir, C. 2020a. *Changes in assortative matching: Theory and evidence for the us* (No. w26932). National Bureau of Economic Research.
- Chiappori, P. A., Costa-Dias, M., Crossman, S., and Meghir, C. 2020b. Changes in assortative matching and inequality in income: Evidence for the UK. *Fiscal Studies*, 41(1), 39-63.
- Chidambaram, B., and Scheiner, J. 2020. Understanding relative commuting within dualearner couples in Germany. *Transportation Research Part A: Policy and Practice*, 134, 113-129.
- Cosaert, S., Theloudis, A., and Verheyden, B. 2023. Togetherness in the Household. *American Economic Journal: Microeconomics*, 15(1), 529-579.
- Davis, J.S., 1993. The commuting of exurban home buyers. *Urban Geography*, 14 (1), 7–29.
- DeSalvo, J. S., and Huq, M. 1996. Income, residential location, and mode choice. *Journal of urban economics*, 40(1), 84-99.
- Dickerson, A., Hole, A. R., and Munford, L. A. 2014. The relationship between well-being and commuting revisited: Does the choice of methodology matter?. *Regional Science and Urban Economics*, 49, 321-329.
- Echeverría, L., Gimenez-Nadal, J. I., and Molina, J. A. 2022. Green mobility and well-being. *Ecological Economics*, 195, 107368. 57: 348–368.

- Eika, L., Mogstad, M., and Zafar, B. 2019. Educational assortative mating and household income inequality. *Journal of Political Economy*, 127(6), 2795-2835.
- Eurostat, 2020. EU Labour Force Survey ad hoc module 2019 on work organization and working time arrangements Luxembourg: Publications Office of the European Union.
- Fan, Y. 2017. Household structure and gender differences in travel time: spouse/partner presence, parenthood, and breadwinner status. *Transportation* 44, 271–291
- Fisher, K., Gershuny, J., Flood, S. M., Backman, D., and Hofferth, S. L. 2019. Multinational Time Use Study Extract System: Version 1.3 [dataset]. Minneapolis, MN: IPUMS.
- Garreau J, 1991. Edge City: Life on the New Frontier. Doubleday, New York
- Gimenez-Nadal, J. I. and Sevilla, A. 2012. Trends in Time Allocation: A Cross-Country Analysis, *European Economic Review*, 56(2), 1338–1359.
- Gimenez-Nadal, J. I., and Molina, J. A. 2014. Commuting time and labour supply in the Netherlands: A time use study. *Journal of Transport Economics and Policy*, 48(3), 409-426.
- Gimenez-Nadal, J. I., and Molina, J. A. 2016. Commuting time and household responsibilities: Evidence using propensity score matching. *Journal of Regional Science*, 56(2), 332-359.
- Gimenez-Nadal, J. I., and Molina, J. A. 2019. Green commuting and gasoline taxes in the United States. *Energy Policy*, *132*, 324-331.
- Gimenez-Nadal, J. I., Molina, J. A., and Velilla, J. 2018a. The commuting behavior of workers in the United States: Differences between the employed and the self-employed. *Journal of Transport Geography*, 66(1), 19-29.
- Gimenez-Nadal, J. I., Molina, J. A., and Velilla, J. 2018b. Spatial distribution of US employment in an urban efficiency wage setting. *Journal of Regional Science*, 58(1), 141-158.
- Giménez-Nadal, J. I., Molina, J. A., and Velilla, J. 2021. Testing urban efficiency wages in France and Spain. *Empirical Economics*, *61*, 2205-2236.

- Giménez-Nadal, J. I., Molina, J. A., and Velilla, J. 2022a. Trends in commuting time of European workers: A cross-country analysis. *Transport Policy*, 116, 327-342.
- Giménez-Nadal, J.I., Molina, J.A., and Velilla. J. 2022b. Intermediate activities while commuting. GLO Discussion Paper, No. 1080
- Giménez-Nadal, J.I., Molina, J.A., and Velilla. J. 2023. Should we cheer together? Gender differences in joint and solo activities' instantaneous well-being, with an application to COVID-19 lockdowns. *Journal of Happiness Studies* 24: 529–562
- Gottholmseder, G., Nowotny, K., Pruckner, G. J., and Theurl, E. 2009. Stress perception and commuting. *Health economics*, 18(5), 559-576.
- Green, A.E., 1997. A question of compromise? Case study evidence on the location and mobility strategies of dual career households. *Regional Studies*, 31 (7), 641–657.
- Gunderson, M., and Oreopolous, P. 2020. Returns to education in developed countries. In *The economics of education* (pp. 39-51). Academic Press.
- Gutierrez-i-Puigarnau, E., van Ommeren, J.N., 2010. Labour supply and commuting. *Journal of Urban Economics*. 68, 82–89.
- Hallberg, D. 2003. Synchronous leisure, jointness and household labor supply. *Labour Economics*, 10(2), 185-203.
- Hamermesh, D. S. 2002. Timing, togetherness and time windfalls. *Journal of population Economics*, 15(4), 601-623.
- Hamermesh, D. S. 2020. Life satisfaction, loneliness and togetherness, with an application to Covid-19 lock-downs. *Review of Economics of the Household*, *18*(4), 983-1000.
- Hanson, S. and Johnston, I. 1985 Gender differences in work-trip length: explanations and implications. Urban Geography, 6, 193–219
- Hanson, S., and Hanson, P., 1980. Gender and urban activity patterns in Uppsala, Sweden. *Geographical Review*, 70 (3), 291–299.
- Hanson, S., and Pratt, G. 1995. Gender, work, and space. London: Routledge Kegan Paul.

- Hansson, E., Mattisson, K., Björk, J., Östergren, P. O., and Jakobsson, K. 2011. Relationship between commuting and health outcomes in a cross-sectional population survey in southern Sweden. *BMC public health*, *11*(1), 1-14.
- Heckman, J. J., Lochner, L. J., and Todd, P. E. 2006. Earnings functions, rates of return and treatment effects: The Mincer equation and beyond. *Handbook of the Economics of Education*, *1*, 307-458.
- Hirte, G., and Illmann, U. 2019. Household decision making on commuting and the commuting paradox. *Empirica*, 46(1), 63–101.
- Jara-Díaz, S., and Rosales-Salas, J. 2015. Understanding time use: Daily or weekly data? Transportation Research Part A: Police and Practice, 76, 38-57.
- Jenkins, S. P., and Osberg, L. 2004. Nobody to play with? The implications of leisure coordination. *Contributions to Economic Analysis*, *271*, 113-145.
- Johnston-Anumonwo, I. 1992. The influence of household type on gender differences in work trip distance. *The Professional Geographer*, 44(2), 161-169.
- Juster, T., and Stafford, F. 1985. *Time, Goods, and Well-Being*. Ann Arbor: Institute for Social Research, The University of Michigan.
- Kim, S., 1994. Gender differences in commuting: an empirical study of the greater Los Angeles metropolitan area. University of California at Irvine, Institute for Transportation Studies, report number UCI-ITS-WP-94-2.
- Kim, S., 1995. Excess commuting for two-worker households in the Los Angeles metropolitan area. *Journal of Urban Economics*, 38, 166–182.
- Knox, P. L., and McCarthy, L. 2005. *Urbanization: an introduction to urban geography*. Upper Saddle River, NJ: Pearson Prentice Hall.
- Künn-Nelen, A. 2016. Does commuting affect health?. *Health Economics*, 25(8), 984-1004.
- Kwon, K., and Akar, G. 2021. Have the gender differences in commuting been shrinking or persistent? Evidence from two-earner households in the US. *International Journal of Sustainable Transportation*, 1-10.

- Lee, B. S., and McDonald, J. F. 2003. Determinants of commuting time and distance for Seoul residents: The impact of family status on the commuting of women. *Urban Studies*, 40(7), 1283-1302.
- Lundberg, S., and Pollak, R. A. 2008. Family decision making (pp. 2060-2066). Palgrave Macmillan UK.
- MacDonald, H. I. 1999. Women's employment and commuting: explaining the links. *Journal of Planning Literature*, *13*(3), 267-283.
- Madden, J. 1981 Why women work closer to home. Urban Studies, 18, 181–94.
- Matthies, E., Kuhn, S., and Klöckner, C. A. 2002. Travel mode choice of women: the result of limitation, ecological norm or weak habit? *Environment and Behavior*, 34(2), 163e177.
- McQuaid, R. W., and Chen, T. 2012. Commuting times—The role of gender, children and part-time work. *Research in transportation economics*, *34*(1), 66-73.
- Mills, E. S. 1967. An aggregative model of resource allocation in a metropolitan area. *The American Economic Review*, 57(2), 197-210.
- Mok, D. 2007. Do two-earner households base their choice of residential location on both incomes?. *Urban Studies*, 44(4), 723-750.
- Molina, J. A., Giménez-Nadal, J. I., and Velilla, J. 2020. Sustainable commuting: Results from a social approach and international evidence on carpooling. *Sustainability*, *12*(22), 9587.
- Motte-Baumvol, B., Bonin, O., and Belton-Chevallier, L. 2017. Who escort children: mum or dad? Exploring gender differences in escorting mobility among Parisian dual-earner couples. *Transportation*, 44(1), 139-157.
- Muller, P. O. 1981. Contemporary Suburban America. Prentice Hall.
- Muth, R.F., 1969. Cities and Housing. University of Chicago Press, Chicago.

- Novaco R. W., and Gonzalez O. I. 2009 Commuting and well-being. In: Amichai-Hamburger Y (ed) Technology and psychological well-being. Cambridge University Press, Cambridge, U.K., pp 174–205
- Oreffice, S., and Sansone, D. 2022. Commuting to work and gender-conforming social norms: evidence from same-sex couples. IZA DP No. 15332
- Picard, N., De Palma, A., and Dantan, S. 2013. Intra-household discrete choice models of mode choice and residential location. *Intra-household discrete choice models of mode choice and residential location*, 419-446.
- Plaut, P. O. 2006. The intra-household choices regarding commuting and housing. *Transportation Research Part A: Policy and Practice*, 40(7), 561-571.
- Roberts, J., and Taylor, K. 2017. Intra-household commuting choices and local labour markets. *Oxford Economic Papers*, 69(3), 734-757
- Roberts, J., Hodgson, R., and Dolan, P. 2011. "It's driving her mad": Gender differences in the effects of commuting on psychological health. *Journal of health economics*, 30(5), 1064-1076.
- Robinson, J.P., and Godbey, G. 1985. *Time for Life: The Surprising Ways Americans Use Their Time*. Penn State University Press, University Park, Pennsylvania.
- Rouwendal, J. 1999. Spatial job search and commuting distances. *Regional Science and Urban Economics*, 29, 491–517.
- Rouwendal, J., Rietvald, P., 1994. Changes in commuting distances of Dutch households. *Urban Studies*, 31 (9), 1545–1557.
- Sandow, E. and Westin, K. 2010. People's Preferences for Commuting in Sparsely Populated Areas: The Case of Sweden, *Journal of Transport and Land Use*, 2(3), 87–107.
- Scheiner, J. 2014. The gendered complexity of daily life: effects of life-course events on changes in activity entropy and tour complexity over time. *Travel Behaviour and Society*, *I*(3), 91-105.

- Schwanen, T. and Dijst, M., 2002. Travel-time ratios for visits to the workplace: the relationship between commuting time and work duration. *Transport Research Part A* 36, 573–592.
- Sermons, M.W. and Koppelman, F.S. 2001 Representing the differences between female and male commute behavior in residential location choice models. *Journal of Transport Geography*, 9, 101–10.
- Singell, L. S. and Lillydahl, J. H. 1986. An Empirical Analysis of the Commute to Work Patterns of Males and Females in Two-Earner Households, *Urban Studies*, 23(2), 119–129.
- Stutzer, A., and Frey, B. S. 2008. Stress that doesn't pay: The commuting paradox. *Scandinavian Journal of Economics*, 110(2), 339-366.
- Sultana, S., 2005. Effects of married-couple dual-earner households on metropolitan commuting: evidence from the Atlanta metropolitan area. *Urban Geography*, 26 (4), 328–352.
- Surprenant-Legault, J., Patterson, Z., and El-Geneidy, A. M. 2013. Commuting trade-offs and distance reduction in two-worker households. *Transportation Research Part A: Policy and Practice*, *51*, 12-28.
- Turner, T. and Neimeier. D. 1997. Travel to Work and Household Responsibility: New Evidence, *Transportation*, 24(4), 397–419.
- Van den Berg, G. J., and Gorter, C. 1997. Job search and commuting time. *Journal of Business & Economic Statistics*, 15(2), 269-281.
- Van Ommeren, J. 1998. On-the-job search behavior: the importance of commuting time. *Land Economics*, 526-540.
- Waldfogel, J. 2007. Parental Work Arrangements and Child Development, *Canadian Public Policy/Analyse de Politiques*, 33(2), 251–272.

Wener, R. E., Evans, G. W., Phillips, D., and Nadler, N. 2003. Running for the 7: 45: The effects of public transit improvements on commuter stress. *Transportation*, 30(2), 203-220.

Tables

Table 1. Commuting Time and Mode of Transport by Gender and Country

		Spain						Italy				
	(i) N	(i) Male		(ii) Female		(iii) = (i)-(ii)		(i) Male		male	(iii) = (i)-(ii)	
	Mean	Std. Dev.	Mean	Std. Dev.	Mean Diff.	Std. Err.	Mean	Std. Dev.	Mean	Std. Dev.	Mean Diff.	Std. Err.
commuting time (minutes)	59.6	40.2	52.4	35.9	7.1***	1.4	60.9	41.0	52.3	37.4	8.6***	1.1
proportion by private transport	0.74	0.42	0.56	0.48	0.18***	0.02	0.83	0.35	0.73	0.41	0.10***	0.01
proportion by public transport	0.09	0.26	0.14	0.32	-0.05***	0.01	0.04	0.16	0.06	0.22	-0.03***	0.01
proportion by active transport	0.18	0.35	0.30	0.43	-0.12***	0.01	0.13	0.30	0.20	0.36	-0.07***	0.01
number of couples		1,447					2,475					

		Korea						United Kingdom					
	(i) N	(i) Male (ii) Female			(iii) = (i	(iii) = (i)-(ii)		(i) Male		(ii) Female		(iii) = (i)-(ii)	
	Mean	Aean Std.		Std.	Mean	Std.	Mean	Std.	Mean	Std.	Mean	Std.	
	Mean	Dev.	Mean	Dev.	Diff.	Err.	Mean	Dev.	Mean	Dev.	Diff.	Err.	
commuting time (minutes)	65.6	38.4	61.0	34.4	4.6***	1.2	65.0	56.0	47.1	37.2	17.9***	2.1	
proportion by private transport	0.74	0.41	0.44	0.46	0.29***	0.01	0.80	0.38	0.73	0.42	0.07***	0.02	
proportion by public transport	0.05	0.18	0.14	0.29	-0.09***	0.01	0.06	0.22	0.07	0.22	-0.01	0.01	
proportion by active transport	0.21	0.37	0.42	0.44	-0.20***	0.01	0.14	0.32	0.20	0.36	-0.06***	0.01	
number of couples		1,737					1,026						

Note: Sample consists of double-earner couples with positive daily commuting times, from the Multinational Time Use Study (MTUS) from 2000 to 2015. Composition of the sample by country is detailed in Table A.1 of the Appendix. Column (iii) reports the difference in the average of commuting times and individual characteristics by gender, using t-type tests on the equality of means.

^{*} Significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level.

Table 2. Descriptive Statistics by Gender and Country

		Spa	ain			It	aly			Ko	rea			U	K	
	M	ale	Femal	e	M	ale	Fen	ıale	Ma	ale	Fen	nale	Ma	ale	Fen	nale
	Mean	Std. Dev.	Mean	Std. Dev.	Mean	Std. Dev.	Mean	Std. Dev.	Mean	Std. Dev.	Mean	Std. Dev.	Mean	Std. Dev.	Mean	Std. Dev.
Panel (A): individual characteristics	_															
age	43.3	8.8	41.1	8.3	44.3	8.4	41.2	8.2	45.8	8.3	42.8	7.9	42.7	10.1	40.6	9.9
elementary education	0.13	0.33	0.11	0.31	0.20	0.40	0.15	0.36	0.15	0.36	0.20	0.40	0.20	0.40	0.18	0.38
secondary education	0.33	0.47	0.29	0.46	0.66	0.47	0.69	0.46	0.58	0.49	0.64	0.48	0.40	0.49	0.39	0.49
higher education	0.54	0.50	0.60	0.49	0.14	0.35	0.16	0.36	0.27	0.44	0.16	0.37	0.40	0.49	0.43	0.50
hrs. of work per week	41.7	8.7	35.8	9.8	-	-	-	-	53.8	15.7	46.7	17.0	44.7	13.7	33.1	14.1
Panel (B): number of children	_															
0	0.40	0.49	0.40	0.49	0.38	0.49	0.38	0.49	0.39	0.49	0.39	0.49	0.50	0.50	0.50	0.50
1	0.31	0.46	0.31	0.46	0.34	0.48	0.34	0.48	0.21	0.41	0.21	0.41	0.21	0.41	0.21	0.41
2	0.26	0.44	0.26	0.44	0.23	0.42	0.23	0.42	0.35	0.48	0.35	0.48	0.24	0.43	0.24	0.43
more than 2	0.04	0.19	0.04	0.19	0.04	0.19	0.04	0.19	0.05	0.22	0.05	0.22	0.06	0.23	0.06	0.23
Panel (C): household characteristics																
urban location	0.72	0.45	0.72	0.45	0.62	0.48	0.62	0.48	0.94	0.25	0.94	0.25	-	-	-	-
owner of a house	-	-	-	-	0.78	0.42	0.78	0.42	0.68	0.47	0.68	0.47	0.84	0.37	0.84	0.37
owner of a vehicle	-	-	-	-	0.99	0.10	0.99	0.10	-	-	-	-	0.60	0.49	0.60	0.49
number of couples		1,4	47			2,	1 75			1,7	737			1,0	026	

Note: Sample consists of double-earner couples with positive daily commuting times, from the Multinational Time Use Study (MTUS) from 2000 to 2015. Composition of the sample by country is detailed in Table A.1 of the Appendix. Information on working hours is not available for Italy (2002-2003); ownership of a home and motorized vehicles is not available for Spain (2009-2010); ownership of motorized vehicles is not available for Korea, and urban location is not available for the UK (2014-2015).

 Table 3. OLS Regressions of Commuting Time, by country

	Sp	oain	It	aly	Ko	rea	United	Kingdom
	Male	Female	Male	Female	Male	Female	Male	Female
M age	-0.017	0.007	0.017	-0.011	0.020	-0.017	0.049	0.000
	(0.026)	(0.027)	(0.022)	(0.020)	(0.025)	(0.023)	(0.031)	(0.029)
M age squared	0.000	-0.000	-0.000	0.000	-0.000	0.000	-0.001	-0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
M secondary education	-0.122	-0.023	0.001	0.014	0.032	0.041	-0.048	0.020
	(0.074)	(0.068)	(0.040)	(0.039)	(0.046)	(0.046)	(0.068)	(0.066)
M higher education	-0.131*	-0.034	0.069	-0.054	0.074	-0.001	0.139*	0.019
	(0.079)	(0.068)	(0.059)	(0.057)	(0.058)	(0.056)	(0.075)	(0.072)
M hours of paid work per week	-0.003	-0.005**	-	-	-0.004***	0.000	-0.001	-0.002
	(0.002)	(0.002)	-	-	(0.001)	(0.001)	(0.002)	(0.002)
F commuting time (log)	0.198***	-	0.203***	-	0.394***	-	0.065*	-
	(0.028)	-	(0.022)	-	(0.025)	-	(0.034)	-
F age	0.052*	-0.040	-0.021	0.004	0.001	0.026	-0.010	-0.020
	(0.028)	(0.028)	(0.022)	(0.020)	(0.027)	(0.022)	(0.031)	(0.030)
F age squared	-0.001*	0.000	0.000	-0.000	-0.000	-0.000	0.000	0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
F secondary education	0.074	-0.044	0.013	-0.004	0.022	-0.058	0.077	0.032
	(0.076)	(0.076)	(0.044)	(0.043)	(0.047)	(0.046)	(0.071)	(0.067)
F higher education	0.043	-0.034	-0.061	0.144**	-0.003	0.019	0.120	0.134*
	(0.083)	(0.079)	(0.058)	(0.057)	(0.064)	(0.062)	(0.082)	(0.073)
F hours of paid work per week	-0.005**	0.008***	-	-	0.001	-0.000	-0.000	0.003*
	(0.002)	(0.002)	-	-	(0.001)	(0.001)	(0.002)	(0.002)
M commuting time (log)	-	0.198***	-	0.191***	-	0.362***	-	0.056*
	-	(0.027)	-	(0.021)	-	(0.024)	-	(0.030)
couples with 1 child (ref.: none)	-0.050	-0.122***	0.046	-0.112***	0.038	0.010	0.091	-0.152**
	(0.046)	(0.044)	(0.032)	(0.032)	(0.039)	(0.037)	(0.065)	(0.063)
couples with 2 children (ref.: none)	-0.038	-0.226***	0.039	-0.218***	-0.007	-0.048	-0.041	-0.192***
- ` '								

	(0.053)	(0.053)	(0.037)	(0.035)	(0.038)	(0.035)	(0.072)	(0.065)
couples with + 2 children (ref.: none)	-0.069	-0.274***	0.158**	-0.325***	-0.048	-0.146**	0.080	-0.133
	(0.092)	(0.106)	(0.073)	(0.068)	(0.068)	(0.062)	(0.113)	(0.094)
urban location	0.005	0.054	0.006	-0.004	0.231**	0.115	-	-
	(0.042)	(0.043)	(0.027)	(0.025)	(0.096)	(0.119)	-	-
owner of a house	-	-	-0.040	-0.020	0.020	-0.020	0.060	0.082
	-	-	(0.032)	(0.031)	(0.029)	(0.027)	(0.070)	(0.068)
owner of a motorized vehicle	-	-	0.049	0.108	-	-	-0.061	-0.190*
	-	-	(0.142)	(0.129)	-	-	(0.140)	(0.102)
constant	3.884***	3.481***	3.275***	3.257***	1.860***	2.145***	2.641***	4.093***
	(0.448)	(0.475)	(0.372)	(0.356)	(0.345)	(0.363)	(0.512)	(0.423)
Occupation controls	Y	res	Y	es	Y	es	1	No
Region controls	Y	es	1	No	N	Го	1	No
Year controls	Y	es	Y	es	Y	es	Y	es
R-squared	0.121	0.162	0.058	0.076	0.261	0.204	0.033	0.058
Number of couples	1,	447	2,	475	1,7	'37	1,	026

Note: Sample consists of double-earner couples with positive daily commuting times, from the Multinational Time Use Study (MTUS) from 2000 to 2015 (see Table A.1 of the Appendix). OLS regressions. Robust standard errors in parentheses. M- Male; F- Female.

* Significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level.

 Table 4. OLS Regressions of the Proportions of Commuting Time by Private Transport, by country

	Sp	ain	It	aly	Ko	rea	United 1	Kingdom
	Male	Female	Male	Female	Male	Female	Male	Female
M age	-0.020	-0.007	-0.001	-0.003	0.031**	-0.016	0.005	0.011
	(0.013)	(0.014)	(0.007)	(0.009)	(0.014)	(0.013)	(0.010)	(0.012)
M age squared	0.000	0.000	0.000	0.000	-0.000**	0.000	-0.000	-0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
M secondary education	0.010	0.009	0.011	0.030*	0.029	-0.008	0.012	0.020
	(0.030)	(0.034)	(0.014)	(0.018)	(0.025)	(0.026)	(0.021)	(0.026)
M higher education	0.019	-0.011	-0.009	0.031	0.023	0.002	-0.001	0.046*
	(0.030)	(0.035)	(0.021)	(0.024)	(0.030)	(0.032)	(0.023)	(0.028)
M hours of paid work per week	0.001	-0.002	-	-	-0.001	0.001	0.001*	0.000
	(0.001)	(0.001)	-	-	(0.000)	(0.000)	(0.001)	(0.001)
F proportion of commuting by private	0.187***	-	0.228***	-	0.266***	-	0.143***	-
	(0.025)	-	(0.020)	-	(0.020)	-	(0.032)	-
F age	0.020	-0.006	-0.004	-0.001	0.002	0.023*	-0.001	-0.017
	(0.014)	(0.014)	(0.007)	(0.008)	(0.014)	(0.012)	(0.010)	(0.011)
F age squared	-0.000	0.000	0.000	-0.000	-0.000	-0.000*	0.000	0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
F secondary education	0.036	-0.020	0.021	-0.015	-0.019	0.038	0.049**	0.043
	(0.033)	(0.038)	(0.016)	(0.019)	(0.024)	(0.026)	(0.023)	(0.028)
F higher education	0.013	0.000	0.004	-0.030	-0.052*	0.155***	-0.011	0.043
	(0.035)	(0.040)	(0.022)	(0.024)	(0.031)	(0.035)	(0.027)	(0.030)
F hours of paid work per week	-0.001	0.003***	-	-	-0.000	0.001**	0.000	0.001**
	(0.001)	(0.001)	-	-	(0.000)	(0.000)	(0.001)	(0.001)
M proportion of commuting by private	-	0.234***	-	0.320***	-	0.347***	-	0.172***
	-	(0.030)	-	(0.027)	-	(0.025)	-	(0.038)
couples with 1 child (ref.: none)	0.030	0.026	0.000	0.042***	0.013	0.007	-0.013	0.037
	(0.020)	(0.022)	(0.012)	(0.014)	(0.018)	(0.021)	(0.022)	(0.024)
couples with 2 children (ref.: none)	0.029	0.028	-0.008	0.047***	-0.002	0.017	0.001	0.025

	(0.022)	(0.024)	(0.013)	(0.016)	(0.018)	(0.020)	(0.023)	(0.025)
couples with + 2 children (ref.: none)	0.026	0.041	0.019	-0.015	0.017	0.054	0.053	0.072**
	(0.039)	(0.045)	(0.023)	(0.031)	(0.030)	(0.036)	(0.033)	(0.037)
urban location	-0.030*	-0.072***	-0.011	-0.003	0.063	-0.090	-	-
	(0.017)	(0.020)	(0.009)	(0.011)	(0.061)	(0.070)	-	-
owner of a house	-	-	0.027**	0.033**	0.043***	0.040**	0.061**	0.082***
	-	-	(0.012)	(0.014)	(0.014)	(0.016)	(0.025)	(0.027)
owner of a motorized vehicle	-	-	0.198***	0.137**	-	-	0.259***	0.204***
	-	-	(0.058)	(0.055)	-	-	(0.055)	(0.053)
constant	-0.091	1.076***	0.398***	0.260*	-0.155	-0.276	0.004	0.067
	(0.194)	(0.217)	(0.133)	(0.156)	(0.233)	(0.200)	(0.166)	(0.175)
Occupation controls	Ŋ	Yes	Y	es	Y	es	Ŋ	No
Region controls	7	Yes	N	Го	N	o	N	No
Year controls	Ŋ	Yes	Y	es	Y	es	Y	es
R-squared	0.094	0.132	0.112	0.121	0.164	0.168	0.113	0.103
Number of couples	1,	447	2,4	175	1,7	37	1,0	026

Note: Sample consists of double-earner couples with positive daily commuting times, from the Multinational Time Use Study (MTUS) from 2000 to 2015 (see Table A.1 of the Appendix). OLS regressions. Robust standard errors in parentheses. M- Male; F- Female.

* Significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level.

Table 5. OLS Regressions of the Proportions of Commuting Time by Public Transport, by country

	Sp	ain	It	aly	Ko	rea	United 1	Kingdom
-	Male	Female	Male	Female	Male	Female	Male	Female
M age	0.009	0.001	0.008**	-0.002	-0.006	-0.015	0.015***	0.002
	(0.007)	(0.010)	(0.003)	(0.005)	(0.009)	(0.012)	(0.005)	(0.006)
M age squared	-0.000	-0.000	-0.000**	-0.000	0.000	0.000	-0.000***	-0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
M secondary education	0.015	-0.016	-0.008	-0.005	0.006	0.022	-0.008	-0.034***
	(0.018)	(0.023)	(0.008)	(0.010)	(0.014)	(0.019)	(0.010)	(0.013)
M higher education	-0.000	0.008	-0.001	-0.002	0.003	0.005	0.018	-0.026*
	(0.018)	(0.022)	(0.012)	(0.014)	(0.017)	(0.023)	(0.012)	(0.015)
M hours of paid work per week	-0.001*	0.000	-	-	-0.000	-0.000	-0.001*	-0.001*
	(0.000)	(0.001)	-	_	(0.000)	(0.000)	(0.000)	(0.000)
F proportion of commuting by public	0.156***	-	0.150***	-	0.080***	-	0.228***	-
	(0.032)	-	(0.029)	-	(0.022)	-	(0.050)	-
F age	0.000	-0.005	-0.003	0.003	-0.001	0.017	-0.005	0.001
	(0.008)	(0.010)	(0.004)	(0.004)	(0.009)	(0.012)	(0.005)	(0.006)
F age squared	0.000	0.000	0.000	-0.000	0.000	-0.000	0.000	0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
F secondary education	-0.019	0.031	-0.006	0.001	0.004	-0.010	0.003	-0.013
	(0.020)	(0.025)	(0.008)	(0.011)	(0.014)	(0.019)	(0.010)	(0.013)
F higher education	0.018	-0.027	0.001	0.013	0.004	-0.042*	0.031**	-0.005
	(0.021)	(0.025)	(0.012)	(0.014)	(0.017)	(0.025)	(0.013)	(0.015)
F hours of paid work per week	-0.000	-0.000	-	_	0.000	-0.001*	-0.001	0.001**
	(0.001)	(0.001)	-	-	(0.000)	(0.000)	(0.000)	(0.000)
M proportion of commuting by public	-	0.224***	-	0.259***	=	0.199***	_	0.244***
	-	(0.043)	-	(0.046)	-	(0.052)	-	(0.050)
couples with 1 child (ref.: none)	-0.017	-0.010	0.001	-0.016*	-0.000	-0.009	0.011	-0.011
	(0.013)	(0.015)	(0.006)	(0.008)	(0.009)	(0.016)	(0.014)	(0.015)
couples with 2 children (ref.: none)	-0.024*	-0.011	-0.004	-0.031***	-0.008	-0.040***	-0.001	-0.009

	(0.013)	(0.016)	(0.007)	(0.009)	(0.009)	(0.014)	(0.015)	(0.015)
couples with + 2 children (ref.: none)	-0.008	-0.006	-0.006	-0.027	-0.005	-0.054**	-0.020	-0.036**
	(0.028)	(0.032)	(0.011)	(0.016)	(0.015)	(0.023)	(0.019)	(0.014)
urban location	0.017*	0.059***	0.007	0.016***	0.071**	0.034	-	-
	(0.009)	(0.011)	(0.005)	(0.006)	(0.031)	(0.039)	-	-
owner of a house	-	-	-0.016**	-0.005	-0.011	-0.009	0.012	-0.033*
	-	-	(0.006)	(0.008)	(0.008)	(0.012)	(0.014)	(0.017)
owner of a motorized vehicle	-	-	-0.092**	-0.027	-	-	-0.119***	-0.035
	-	-	(0.044)	(0.037)	-	-	(0.042)	(0.033)
constant	0.377***	-0.215	-0.018	0.076	0.059	-0.072	-0.010	0.119
	(0.112)	(0.144)	(0.066)	(0.086)	(0.087)	(0.140)	(0.095)	(0.103)
Occupation controls	Y	es	Y	es	Y	es	N	О
Region controls	Y	es	N	lo .	N	lo	N	o
Year controls	Y	es	Y	es	Y	es	Y	es
R-squared	0.158	0.189	0.076	0.081	0.044	0.051	0.116	0.110
Number of couples	1,4	447	2,4	475	1,7	737	1,0	26

Note: Sample consists of double-earner couples with positive daily commuting times, from the Multinational Time Use Study (MTUS) from 2000 to 2015 (see Table A.1 of the Appendix). OLS regressions. Robust standard errors in parentheses. M- Male; F- Female.

* Significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level.

 Table 6. OLS Regressions of the Proportions of Commuting Time by Active Transport, by country

	Sp	ain	It	aly	Ko	rea	United 1	Kingdom
·	Male	Female	Male	Female	Male	Female	Male	Female
M age	0.012	0.006	-0.006	0.005	-0.026**	0.028**	-0.020**	-0.015
	(0.011)	(0.012)	(0.007)	(0.008)	(0.012)	(0.014)	(0.009)	(0.011)
M age squared	-0.000	-0.000	0.000	-0.000	0.000**	-0.000*	0.000*	0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
M secondary education	-0.031	0.012	-0.001	-0.027*	-0.028	-0.008	0.001	0.011
	(0.027)	(0.033)	(0.013)	(0.016)	(0.023)	(0.025)	(0.019)	(0.024)
M higher education	-0.023	0.005	0.018	-0.033	-0.021	-0.004	-0.014	-0.024
	(0.027)	(0.034)	(0.019)	(0.022)	(0.027)	(0.032)	(0.020)	(0.024)
M hours of paid work per week	-0.000	0.001	_	-	0.001*	-0.001	-0.001	0.000
	(0.001)	(0.001)	-	-	(0.000)	(0.000)	(0.000)	(0.001)
F proportion of commuting by active	0.177***	_	0.252***	-	0.294***	-	0.102***	-
	(0.025)	-	(0.021)	-	(0.020)	-	(0.032)	-
F age	-0.022*	0.012	0.009	-0.003	-0.000	-0.035**	0.008	0.017*
	(0.011)	(0.012)	(0.006)	(0.007)	(0.012)	(0.014)	(0.009)	(0.010)
F age squared	0.000*	-0.000	-0.000	0.000	-0.000	0.000**	-0.000	-0.000*
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
F secondary education	-0.014	-0.015	-0.014	0.019	0.012	-0.027	-0.051**	-0.040
	(0.030)	(0.036)	(0.015)	(0.017)	(0.022)	(0.025)	(0.022)	(0.026)
F higher education	-0.031	0.022	-0.006	0.030	0.043	-0.117***	-0.014	-0.044
	(0.031)	(0.038)	(0.020)	(0.022)	(0.028)	(0.034)	(0.024)	(0.027)
F hours of paid work per week	0.001	-0.002***	_	-	0.000	-0.000	-0.000	-0.002***
	(0.001)	(0.001)	-	-	(0.000)	(0.000)	(0.000)	(0.001)
M proportion of commuting by active	-	0.251***	-	0.345***	-	0.426***	-	0.130***
	-	(0.035)	-	(0.028)	-	(0.026)	-	(0.041)
couples with 1 child (ref.: none)	-0.015	-0.018	-0.004	-0.025**	-0.018	0.005	0.004	-0.030
	(0.016)	(0.019)	(0.010)	(0.012)	(0.017)	(0.020)	(0.019)	(0.021)
couples with 2 children (ref.: none)	-0.008	-0.020	0.009	-0.019	-0.000	0.016	0.003	-0.015

	(0.019)	(0.022)	(0.012)	(0.014)	(0.016)	(0.019)	(0.019)	(0.022)
couples with + 2 children (ref.: none)	-0.010	-0.043	-0.012	0.037	-0.024	-0.007	-0.038	-0.039
	(0.032)	(0.041)	(0.021)	(0.029)	(0.027)	(0.034)	(0.028)	(0.035)
urban location	0.015	0.019	0.006	-0.008	-0.129**	0.064	-	-
	(0.016)	(0.019)	(0.009)	(0.010)	(0.056)	(0.067)	-	-
owner of a house	-	-	-0.013	-0.029**	-0.036***	-0.029*	-0.074***	-0.057**
	-	-	(0.011)	(0.012)	(0.013)	(0.015)	(0.023)	(0.024)
owner of a motorized vehicle	-	-	-0.112**	-0.102*	-	-	-0.143***	-0.171***
	-	-	(0.055)	(0.056)	-	-	(0.054)	(0.053)
constant	0.355**	-0.385*	0.130	0.150	0.576***	0.744***	0.590***	0.472***
	(0.169)	(0.202)	(0.126)	(0.144)	(0.219)	(0.192)	(0.158)	(0.164)
Occupation controls	Y	es	Y	es	Y	es	N	lo
Region controls	Y	es	N	lo	N	lo	N	lo .
Year controls	Y	es	Y	es	Y	es	Y	es
R-squared	0.102	0.125	0.122	0.143	0.205	0.185	0.081	0.092
Number of couples	1,4	147	2,4	175	1,7	737	1,0	026

Note: Sample consists of double-earner couples with positive daily commuting times, from the Multinational Time Use Study (MTUS) from 2000 to 2015 (see Table A.1 of the Appendix). OLS regressions. Robust standard errors in parentheses. M- Male; F- Female.

* Significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level.

Table 7. Multinomial Logit Regressions of Commuting Mode Choice, by country

	Sp	ain	Spa	ain	Ita	aly	It	aly
	M	ale	Fen	nale	M	ale	Fei	nale
	Public	Active	Public	Active	Public	Active	Public	Active
M age	0.252*	0.153	0.007	0.070	0.356*	-0.061	0.105	0.011
	(0.153)	(0.148)	(0.141)	(0.113)	(0.194)	(0.104)	(0.135)	(0.089)
M age squared	-0.003*	-0.002	-0.000	-0.001	-0.004*	0.001	-0.002	-0.000
	(0.002)	(0.002)	(0.002)	(0.001)	(0.002)	(0.001)	(0.002)	(0.001)
M secondary education	0.061	-0.277	-0.337	0.033	-0.476	-0.163	-0.147	-0.301*
	(0.419)	(0.305)	(0.366)	(0.258)	(0.347)	(0.194)	(0.305)	(0.171)
M higher education	-0.109	-0.225	0.068	-0.018	-0.346	0.056	0.046	-0.335
	(0.452)	(0.310)	(0.374)	(0.263)	(0.462)	(0.289)	(0.368)	(0.267)
M hours of paid work per week	-0.027	0.001	-0.001	0.008	-	-	-	-
	(0.019)	(0.010)	(0.013)	(0.009)	-	-	-	-
F age	-0.062	-0.191	-0.050	0.057	-0.137	0.103	-0.004	0.073
	(0.160)	(0.139)	(0.142)	(0.115)	(0.169)	(0.103)	(0.120)	(0.090)
F age squared	0.001	0.002	0.001	-0.000	0.002	-0.001	0.001	-0.000
	(0.002)	(0.002)	(0.002)	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)
F secondary education	-0.354	-0.231	0.295	-0.056	-0.450	-0.249	-0.187	0.117
	(0.438)	(0.316)	(0.401)	(0.267)	(0.364)	(0.222)	(0.324)	(0.186)
F higher education	0.206	-0.337	-0.405	-0.201	-0.024	0.035	0.179	0.166
-	(0.420)	(0.312)	(0.395)	(0.261)	(0.491)	(0.313)	(0.383)	(0.273)
F hours of paid work per week	-0.009	0.013	-0.017*	-0.014*	-	-	-	-
•	(0.013)	(0.008)	(0.010)	(0.007)	-	-	-	-
couples with 1 child (ref.: none)	-0.355	-0.237	-0.291	-0.146	-0.132	-0.033	-0.523**	-0.360**
, , ,	(0.259)	(0.204)	(0.224)	(0.166)	(0.275)	(0.170)	(0.210)	(0.144)
couples with 2 children (ref.: none)	-0.515*	-0.147	-0.286	-0.116	-0.369	0.119	-0.881***	-0.192
, ,	(0.290)	(0.220)	(0.256)	(0.184)	(0.337)	(0.194)	(0.259)	(0.165)
couples with + 2 children (ref.: none)	-0.080	0.076	-0.312	-0.356	-0.471	-0.214	-0.539	0.403
	(0.486)	(0.411)	(0.464)	(0.368)	(0.605)	(0.384)	(0.508)	(0.277)
urban location	0.867***	0.215	1.419***	0.269*	0.544**	-0.033	0.271	-0.103
	(0.308)	(0.183)	(0.276)	(0.150)	(0.256)	(0.138)	(0.183)	(0.116)
owner of a house	-	-	-	-	-0.740***	-0.363**	-0.230	-0.421***

	-	-	-	-	(0.241)	(0.160)	(0.195)	(0.134)
owner of a motorized vehicle	-	-	-	-	-2.603***	-1.620***	-1.518**	-1.518***
	-	-	-	-	(0.617)	(0.513)	(0.633)	(0.457)
constant	-6.355**	-1.996	-2.090	-4.390**	-6.579**	-1.640	-2.469	-2.258
	(2.817)	(2.272)	(2.377)	(1.851)	(3.176)	(1.887)	(2.151)	(1.679)
Occupation controls	Y	es	Ŋ	Zes .	Y	es	7	Zes .
Region controls	Y	es	Y	es	N	lo	1	No
Year controls	Y	es	Y	es	Y	es	Ŋ	Zes .
Number of couples	1,4	147	1,	447	2,4	175	2,	475

Note: Sample consists of double-earner couples with positive daily commuting times, from the Multinational Time Use Study (MTUS) from 2000 to 2015 (see Table A.1 of the Appendix). OLS regressions. Robust standard errors in parentheses. M- Male; F- Female.

* Significant at the 10% level; *** significant at the 5% level; *** significant at the 1% level.

 Table 7 (Cont.).
 Multinomial Logit Regressions of Commuting Mode Choice, by country

	Korea Male		Ke	Korea		United Kingdom		United Kingdom	
			Female		Male		Female		
	Public	Active	Public	Active	Public	Active	Public	Active	
M age	-0.177	-0.204*	-0.194	0.130	0.319**	-0.279**	0.211	-0.122	
	(0.306)	(0.111)	(0.178)	(0.102)	(0.152)	(0.115)	(0.182)	(0.112)	
M age squared	0.003	0.002**	0.002	-0.001	-0.004**	0.003**	-0.003	0.002	
	(0.003)	(0.001)	(0.002)	(0.001)	(0.002)	(0.001)	(0.002)	(0.001)	
M secondary education	0.034	-0.392*	0.315	-0.014	-0.565	-0.052	-0.988***	0.104	
	(0.540)	(0.213)	(0.332)	(0.195)	(0.409)	(0.279)	(0.357)	(0.226)	
M higher education	-0.126	-0.338	0.004	-0.043	0.214	-0.239	-0.647*	-0.304	
	(0.621)	(0.255)	(0.394)	(0.239)	(0.379)	(0.316)	(0.392)	(0.267)	
M hours of paid work per week	0.000	0.009**	-0.007	-0.001	-0.015	-0.005	-0.030***	0.005	
	(0.009)	(0.004)	(0.005)	(0.004)	(0.011)	(0.007)	(0.011)	(0.006)	
F age	-0.118	-0.056	0.070	-0.250**	-0.120	0.136	-0.041	0.174	
	(0.299)	(0.114)	(0.188)	(0.099)	(0.145)	(0.137)	(0.141)	(0.115)	
F age squared	0.001	0.001	-0.001	0.003**	0.001	-0.001	0.001	-0.002*	
	(0.003)	(0.001)	(0.002)	(0.001)	(0.002)	(0.002)	(0.002)	(0.001)	
F secondary education	0.279	0.070	-0.203	-0.383**	0.170	-0.735**	-0.349	-0.308	
	(0.542)	(0.226)	(0.296)	(0.192)	(0.437)	(0.290)	(0.431)	(0.231)	
F higher education	0.117	-0.016	-0.838**	-1.218***	1.151**	-0.216	-0.207	-0.506*	
	(0.654)	(0.295)	(0.404)	(0.259)	(0.452)	(0.329)	(0.470)	(0.266)	
F hours of paid work per week	-0.006	0.003	-0.006	-0.006*	-0.013	0.000	0.014*	-0.022***	
	(0.008)	(0.004)	(0.005)	(0.003)	(0.010)	(0.007)	(0.008)	(0.007)	
couples with 1 child (ref.: none)	0.061	-0.214	-0.036	-0.071	0.295	0.063	-0.079	-0.343	
•	(0.328)	(0.193)	(0.224)	(0.161)	(0.356)	(0.275)	(0.372)	(0.240)	
couples with 2 children (ref.: none)	-0.453	0.028	-0.400*	0.013	-0.324	-0.019	-0.176	-0.257	
	(0.355)	(0.179)	(0.223)	(0.155)	(0.422)	(0.290)	(0.350)	(0.243)	
couples with + 2 children (ref.: none)	-0.519	-0.125	-0.887*	-0.097	-1.055	-0.823	-13.806***	-0.575	
*	(0.757)	(0.336)	(0.499)	(0.262)	(0.870)	(0.627)	(0.340)	(0.387)	
urban location	15.426***	-1.006***	2.380**	0.304	-	-	· -	-	
	(0.281)	(0.239)	(0.956)	(0.548)	-	-	-	-	
owner of a house	-0.477	-0.505***	-0.410**	-0.395***	-0.015	-0.925***	-0.632*	-0.636***	

	(0.292)	(0.136)	(0.178)	(0.120)	(0.411)	(0.240)	(0.332)	(0.233)
owner of a motorized vehicle	_	-	-	-	-2.979***	-1.934***	-1.747***	-1.499***
	-	-	-	-	(0.524)	(0.440)	(0.587)	(0.400)
constant	-12.024***	4.910***	-11.778***	1.891	-3.138	4.384**	-1.883	1.069
	(2.612)	(1.574)	(2.211)	(1.802)	(2.989)	(1.960)	(2.904)	(1.704)
Occupation controls	Y	es	Ye	es	N	lo	1	No
Region controls	N	lo	N	О	N	lo	1	No
Year controls	Y	es	Ye	es	Y	es	Ŋ	Zes .
Number of couples	1,7	737	1,7	37	1,0)26	1,	026

Note: Sample consists of double-earner couples with positive daily commuting times, from the Multinational Time Use Study (MTUS) from 2000 to 2015 (see Table A.1 of the Appendix). OLS regressions. Robust standard errors in parentheses. M- Male; F- Female.

* Significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level.

Appendix

Table A.1. Sample Composition

Country	Survey Years	Number of Couples		
Spain	2002-2003 and 2009-2010	1,448		
Italy	2002-2003 and 2008-2009	2,540		
Korea	2009	1,737		
UK	2000-2001 and 2014-2015	1,026		
All countries	2000 to 2015	6,751		

Note: Sample consists of double-earner couples with positive daily commuting times, from the Multinational Time Use Study (MTUS) from 2000 to 2015.